Author's response to reviews

Title: Citrus aurantium L. essential oil exhibits anxiolytic-like activity mediated by 5-HT1A-receptors and reduces cholesterol after repeated oral treatment

Authors:

Celso A R A Costa (celso@ibb.unesp.br)
Thais C Cury (thacury@gmail.com)
Bruna O Cassettari (brucassettari@gmail.com)
Regina K Takahira (takahira@fmvz.unesp.br)
Jorge C Flório (jflorio@usp.br)
Mirtes Costa (farmacologia@ibb.unesp.br)

Version: 4 Date: 4 February 2013

Author's response to reviews: see over
Dear editor,

According to your solicitation we provided modification in the manuscript in order to clearly specify the individual contribution of each author. The following text was inserted in the Authors’ Contributions section:

**Authors’ contributions**

CARAC participated in the design of the study, carried out all the experimental procedures and drafted the manuscript. TCC and BOC participated in experimental protocols and analyzed the behavioral data. RKT made and supervised all biochemical analyzes. JCF participated in the design of the study and made the neurochemical analyses. MC supervised the entire project, participated in its design and statistical analysis and helped to the final form of the manuscript. The work was part of the Ph.D. dissertation of CARAC and the graduation work of TCC and BOC. All of the authors significantly contributed to and have approved the final manuscript.

We hope that in this way the manuscript meets all standards for publication,

Best regards

Sincerely,

Mirtes Costa, Ph.D.
Department of Pharmacology

Institute of Bioscience - São Paulo State University (UNESP)
18618-970- Botucatu, São Paulo, Brazil.
FAX: + (55) 14. 3815 3744
*E-mail address:* farmacologia@ibb.unesp.br
Alternative E-mail address: mirtescosta@uol.com.br